Author's response to reviews

Title: Prevalence of developmental delays in early infancy: results from a regional population-based longitudinal study

Authors:

Lisbeth Valla (Lisbeth.valla@r-bup.no)
Tore Wentzel-Larsen (tore.wentzel-larsen@r-bup.no)
Dag Hofoss (dag.hofoss@medisin.uio)
Kari Slinning (kari.slinning@r-bup.no)

Version: 2 Date: 28 January 2015

Author's response to reviews:

I have Attached a cover letter, describing one change in my article: I have put line numbers in the article.

Regards Lisbeth Valla